Author's response to reviews

Title: Up-to-date on mortality in COPD - report from the OLIN COPD study

Authors:

Anne Lindberg (anne.lindberg@nll.se)
Lars-Gunnar Larsson (lars-gunnar.larsson@nll.se)
Hana Muellerova (hana.x.muellerova@gsk.com)
Eva Rönmark (eva.ronmark@nll.se)
Bo Lundbäck (bo.lundback@gu.se)

Version: 5 Date: 3 January 2012

Author's response to reviews:

Dear Editor,

we sincerely appreciate the positive feedback and apologize for submitting a version not fully conforming to the journal style.

The major revisions regarding affiliations and manuscript structure are made according to recommendation.

The following minor revisions are made according to recommendations: title page structure has been changed, updated affiliations are marked by "current address", the paragraph authors contributions has been revised and the visible vertikal lines are removed from the tables.

Unfortunately I was not able to remove the title in the figure, and I sincerely hope that this change can be made by the editorial office as this was stated as a minor revision.

Yours sincerely

Anne Lindberg, Associate professor, senior lecturer, MD